Author's response to reviews

Title: Dietary intakes and food sources of fatty acids in diets of Guatemalan schoolchildren: A cross-sectional study

Authors:

Odilia I. Bermudez (Odilia.bermudez@tufts.edu)
Claire Toher (Claire_toher@hotmail.com)
Gabriela Montenegro-Bethancourt (cessiam@quate.net.gt)
Marieke Vossenaar (mvossenaar@hotmail.com)
Paul Mathias (Paul.mathias@dit.ie)
Colleen Doak (colleen.doak@falw.vu.nl)
Noel W. Solomons (cessiam@quate.net.gt)

Version: 3 Date: 26 September 2009

Author's response to reviews: see over
September 26, 2009

The Nutrition Journal Editorial Team
Leesburg Regional Medical Center
& Medical Director, MedSource Nutrition Center
Lady Lake, Florida, USA

Dear Editorial Team,

This is to re-submit the revised manuscript “Dietary intakes and food sources of fat and fatty acids in diets of Guatemalan schoolchildren: A cross-sectional study” identified as MS 128681614286857. We ask for your acceptance of this resubmission even that we are almost 1 day past the deadline given to us (September 25th). Unfortunately we have a last minute problem with the online communication between authors that delayed assembling the documents earlier.

After receiving the comments from the reviewers and the journal editors, we did and extensive review with them and took into consideration the valuable comments we received. As a result, we are resubmitting for your consideration the revised manuscript, with the required modifications. The newly revised manuscript is enclosed here (manuscript and figures), plus our rebuttal to the reviewers and editors – also enclosed here.

We appreciate if this revised manuscript is considered for publication after the additional reviews are evaluated. Please let us know if there is anything else we need to take care of.

Thank you very much. We look forward to hearing from you.

Sincerely,

Odilia I. Bermudez, Ph.D., MPH, LDN